Author’s response to reviews

Title: Influence of subinhibitory antifungal concentrations on extracellular hydrolases and biofilm production in Candida albicans recovered from Egyptian patients

Authors:

Houdaii El-Houssaini (houdaii.housam@hu.edu.eg)
Omnia Elnabawy (dr_omniamohamed@yahoo.com)
Hebatallah Nasser (heba.nasser@hu.edu.eg)
Walid Elkhatib (walid-elkhatib@pharma.asu.edu.eg)

Version: 4 Date: 20 Dec 2018

Author’s response to reviews:

1. It has come to our attention that within your manuscript there is significant text overlap with other publications, particularly:

Background:


Methods:


Discussion:


Authors' response: Rephrasing has been considered in the mentioned sections as recommended by the editor in the technical report.

2. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Authors' response: Authors’ names have been mentioned in their full initials in the Authors’ Contributions section as recommended by the editor in the technical report.

3. Reviewer's report concerning English language revision has been considered and unnecessary verbiage has been removed from the manuscript as recommended by reviewer 2.